Author's response to reviews

Title: Air Column in esophagus and Symptoms of Gastroesophageal Reflux Disease

Authors:

Alijavad Mousavi (MousaviAlijavad@yahoo.com)
Hanie Raji (RajiHanie@yahoo.com)
Mojtaba Teimoori (moitaba_teimoori@irimc.ir)
Shadi Ghourchian (shadighourchian@yahoo.com)

Version: 6 Date: 20 December 2011

Author's response to reviews:

1. We stated a 'Competing interests' section between the Conclusions and Authors' contributions. The authors declare that they have no conflict of interests.

Financial competing interests

- In the past five years we have not received reimbursements, fees, funding, or salary from an organization that may have any financial gains from the publication of this manuscript, either in now or in future.
- We do not hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future.
- We do neither hold nor currently applying for any patents relating to the content of the manuscript. We have not received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript.
- We have not any financial competing interests.

Non-financial competing interests

- There are not any non-financial competing interests (political, personal, religious, academic, ideological, intellectual, commercial or any other).